Author's response to reviews

Title: Randomised controlled comparison of the Health Survey Short Form (SF-12) and the Graded Chronic Pain Scale (GCPS) in telephone interviews versus self-administered questionnaires. Are the results equivalent?

Authors:

Margitta Lungenhausen (margitta.lungenhausen@bergmannsheil.de)
Stefan Lange (stefan.lange@iqwig.de)
Christoph Maier (christoph.maier@ruhr-uni-bochum.de)
Claudia Schaub (claudia.schaub@ruhr-uni-bochum.de)
Hans J Trampisch (hans.j.trampisch@ruhr-uni-bochum.de)
Heinz G Endres (heinz.endres@ruhr-uni-bochum.de)

Version: 4 Date: 20 November 2007

Author's response to reviews: see over
Randomised controlled comparison of the Health Survey Short Form (SF-12) and the Graded Chronic Pain Scale (GCPS) in telephone interviews versus self-administered questionnaires.
Are the results equivalent?

Dear Sir or Madam

In the submitted revision please find the requested formatting changes.

With best regards from Germany

Heinz G. Endres
Ruhr-University Bochum